Author's response to reviews

Title: Utility of Total lymphocyte Count as a Surrogate Marker for CD4 counts in HIV-1 infected Children in Kenya

Authors:

Leah Githinji (lgithinji2002@yahoo.com)
Elizabeth Malecha Obimbo (eobimbo@hcc.or.ke)
Thomas Nderitu (tnderitu@yahoo.com)
Dalton C Wamalwa (dalton@africaonline.co.ke)
Dorothy A Mbori-Ngacha (dngacha@gmail.com)

Version: 3 Date: 3 August 2011

Author's response to reviews:

Dear Editor,

Thanks very much once more for the revision to our manuscript. In this version we have addressed all the minor concerns raised by the reviewer. Specifically all typographic errors identified have been corrected accordingly and the citation to PEPFAR 2 goals added.

Thankyou for your consideration.

Dr Dalton Wamalwa